Author’s response to reviews

Title: Sustained serological and complete responses in HBeAg-Positive Patients treated with Peginterferon alfa-2b: a 6-Year Long-Term Follow-Up of a Multicenter, Randomized, Controlled Trial in China

Authors:

Jian Sun (sunjn09@126.com)
Huiguo Ding (dinghuiguo09@163.com)
Guofeng Chen (chengf10@yeah.net)
Guiqiang Wang (kinggqiang06@163.com)
Lai Wei (weilaiwl09@163.com)
Jiming Zhang (zhangjm06@yeah.net)
Qing Xie (drxieqing5@126.com)
Mobin Wan (wanbomin13@126.com)
Hong Tang (tanghongc@yeahmca.cn)
Shijun Chen (cshijun17@yeah.net)
Zhiliang Gao (zhiliang10@mca21cn.cn)
Yuming Wang (kingym11@163.com)
Dazhi Zhang (zhdazhi15@126.com)
Wenxiang Huang (huangwenxiangwx@163.com)
Jifang Sheng (shengjif6@yeahmca.cn)
Qin Ning (ningqean06@163.com)
Dongliang Yang (dongliangy@mca21cn.cn)
Jian Lu (lujianlj09@126.com)
Chen Pan (chenpan16@yeah.net)
Author’s response to reviews:

1. In the “Ethics approval and consent to participate” section of the Declarations please include the FULL name of the ethics committee(s) (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

Thanks for the comment, we already add the details of all ethics approval and consent to participate in the revised manuscript.